%
% Fill only lines marked with dots  
% 
%
%
\documentclass[10pt,a4paper,twoside]{article}
\usepackage{latexsym,amssymb}
\usepackage{amssymb,amsmath}
\usepackage{graphicx}
\usepackage{color}
\usepackage{fancyhdr}
\pagestyle{fancy}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  PLEASE, ONLY IN ENGLISH                                         %
%  PLEASE, ONLY ONE PAGE                                           %
%  PLEASE, USE LATEX2E                                             %
%                                                                  %
%  PLEASE, DO NOT DEFINE NEW COMMANDS                              %
%  USE NEWTHEOREMS BELOW, PLEASE, IF YOU WRITE THEOREMS AND SUCH   % %                                                                  %
%  PLEASE, CONTROL THE SIZE OF YOUR ABSTRACT BEFORE YOU SEND IT    %
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
\newtheorem{thm}{Theorem}
\newtheorem{defi}[thm]{Definition}
\newtheorem{axi}[thm]{Axiom}
\newtheorem{clm}[thm]{Claim}
\newtheorem{col}[thm]{Corollary}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{con}[thm]{Conjecture}
\newtheorem{hyp}[thm]{Hypothesis}
\newtheorem{rem}[thm]{Remark}
\newcommand{\proof}{\noindent{\bf Proof:}\hspace{0.2cm}}
\setcounter{thm}{0}
\newcommand{\N}{{\mathbb N}}
\newcommand{\R}{{\mathbb R}}
\newcommand{\W}{W_0^{1,p}(\Omega)}
\newcommand{\eps}{{\varepsilon}}
\newcommand{\ds}{\displaystyle}

\newcommand{\F}{\mathcal F}
\newcommand{\E}{\mathscr E}
\newcommand{\J}{\mathscr{J}}
\newcommand{\Pl}{(\mathscr{P})}
\newcommand{\CC}{\mathbb C}
\newcommand{\into}{\int\limits_{\Omega}}
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\textwidth 15.0cm
\textheight 22.0cm
\oddsidemargin 0.4cm
\evensidemargin 0.4cm
\topmargin  0.0cm

\frenchspacing

\headwidth=\textwidth 	
\fancyhead[HCO]{9th Int. Conf. on Mathematics and Informatics,
	September 7-8, 2023, T\^argu Mure\c s, Romania}
\lhead{}\rhead{}
\fancyfoot{}
%


\begin{document}
\begin{thebibliography}{9} 
% Please, use the following patterns
 \bibitem{Dunn} J. C. Dunn, A fuzzy relative of the ISODATA process and its use in detecting compact well-separated clusters, \textit{Journal of Cybernetics} \textbf{1}, 3 (1973) 32–-57.
 \bibitem{Bezdek} J. C. Bezdek, R. Ehrlich, and W. Full, Fcm: The fuzzy c-means clustering algorithm, \textit{Computers \& Geosciences} \textbf{10}, 2 (1984) 191–-203.
 \bibitem{Krishna}  R. Krishnapuram and J. Keller, A possibilistic approach to clustering, \textit{IEEE Transactions on Fuzzy Systems} \textbf{1}, 2 (1993) 98-–110.
 \bibitem{Pal}  N. Pal, K. Pal, J. Keller, and J. Bezdek, A possibilistic fuzzy c-means clustering algorithm, \textit{IEEE Transactions on Fuzzy Systems} \textbf{13}, 4 (2005) 517-–530.
 \bibitem{Miyamoto}  S. Miyamoto and N. Kurosawa, Controlling cluster volume sizes in fuzzy c-means clustering, \textit{SCIS and ISIS}, Yokohama, Japan, (2004) 1-–4.
 \bibitem{Lalo}  L. Szilágyi, S. Lefkovits, and S. Szilágyi, Self-tuning possibilistic c-means clustering models, \textit{International Journal of Uncertainty, Fuzziness and Knowledge-Based Systems} \textbf{27}, Suppl. 1 (2019) 143–-159.

\end{thebibliography}


\begin{center}{\Large\bf
An Adaptive Parameter Setting Technique for the Possibilistic Fuzzy $c$-Means Clustering Algorithm
}\end{center}
%
\begin{center}{\large\bf\noindent
 Mirtill Boglárka Naghi
}\\[2mm]
%%
Óbuda University, Budapest, Hungary \\ 
Sapientia Hungarian University of Transylvania, T\^argu Mure\c s, Romania
%%
\\[1mm]\texttt{
%%	
naghi.mirtill@gmail.com
%%
}\end{center}
%

\vspace*{7mm}
% 
$C$-means clustering algorithms employing fuzzy partitions have two basic approaches. Chronologically the first, introduced by Dunn \cite{Dunn} and generalized by Bezdek \cite{Bezdek} uses a probalilistic partition, and is called the fuzzy $c$-means (FCM) clustering model. Alternatively, the so-called possibilistic $c$-means (PCM) algorithm, introduced by Krishnapuram and Keller \cite{Krishna}, relaxed the probabilistic constraint of the partition matrix and uses fuzzy membership functions that describe the compatibility of data vectors with the $c$ clusters. The irony in the terminology is that PCM is more fuzzy than FCM.

Since both basic approaches have shortcomings in their behavior, mixed clustering methods were proposed, which were desingned to attenuate the unwanted phenomena. The most well-known mixed $c$-means clustering algorithm is the possibilistic-fuzzy
$c$-means model proposed by Pal et al. \cite{Pal}, which combines the probabilistic and possibilistic components of the partition matrix as a linear combination.  

This paper presents an alternative formulation of the PFCM clustering model, which allows for the adaptive tuning of the so-called possibilistic penalty terms of the algorithm. This way the user needs to set a reduced number of parameters, which makes the algorithm easier to handle. The proposed solution is inspired by the cluster size controlling version of the FCM algorithm proposed by Miyamoto and Kurosawa \cite{Miyamoto}, and the self-tuning version of the PCM algorithm proposed by Szilágyi et al \cite{Lalo}.
Numerical tests revealed that the proposed adaptive PFCM algorithm produces slightly better clusteing results in terms of partition quality indicators like cluster purity, normalized mutual information, and adjusted Rand index.


%
%

%

\end{document}


